Author's response to reviews

Title: Patient acceptance and perceived utility of pre-consultation prevention summaries and reminders: pilot study

Authors:

Oliver R Frank (oliver.frank@adelaide.edu.au)
Nigel P Stocks (nigel.stocks@adelaide.edu.au)
Paul Aylward (paul.aylward@adelaide.edu.au)

Version: 2 Date: 2 September 2010

Author's response to reviews:

We have changed the name of the fictitious patient in Figure 1 from "Ms. Melinda Baker" to "[patient name]".